RESEARCH

Open Access

The assessment of caregiver self-efficacy in a virtual eating disorder setting



Nickolas M. Jones^{1,2}, Jessica H. Baker¹, Bek Urban¹, David Freestone¹, Angela Celio Doyle¹, Cara Bohon^{1,3} and Dori M. Steinberg^{1,4,5*}

Abstract

Background Caregiver self-efficacy is thought to be a key component for successful family-based treatment (FBT) for individuals with eating disorders. As such, interventions aimed at enhancing caregiver self-efficacy, often measured via the Parents Versus Anorexia scale, have been a focal point of FBT literature. However, studies looking at the relationship between caregiver self-efficacy and treatment outcomes have been mixed. We aimed to better understand the influence of caregiver self-efficacy on eating disorder treatment outcomes during FBT.

Methods Caregiver self-efficacy was measured using the Parents Versus Eating Disorders (PVED) scale, an adapted version of the Parents Versus Anorexia scale, in a sample of 1051 patients with an eating disorder and 1528 caregivers (patients can have more than one caregiver) receiving virtual FBT. Across two multilevel models, we tested how caregiver self-efficacy changed over time and its association with changes in eating disorder symptoms and weight over the first 16 weeks of treatment.

Results Over treatment, PVED scores increased (b = 0.79, SE = 0.04, CI [0.72, 0.86]) and starting PVED scores were predictive of improved eating disorder symptoms (b = -0.73, SE = 0.22, CI [-1.15, -0.30]), but not weight (b = -0.96, SE = 0.59, CI [-2.10, 0.19]). We also found that PVED change-from-baseline scores were predictive of weight (b = -0.48, SE = 0.03, CI [-0.53, -0.43]) such that patient weight was lower when caregiver reports of PVED were higher. Likewise, the association between caregiver change in PVED scores and weight varied as a function of treatment time (b = 0.27, SE = 0.01, CI [0.24, 0.29]). Results were consistent when isolating patients with anorexia nervosa.

Conclusions Caregiver self-efficacy during FBT improved over time but was not robustly associated with treatment outcomes. This may, in part, be due to psychometric properties of the PVED scale. We describe these issues and illustrate the need for development of a new measure of self-efficacy for caregivers supporting their loved ones through eating disorder treatment.

Keywords Eating disorder, Family therapy, Anorexia nervosa, Self efficacy

Plain English summary

In family-based treatment (FBT) for eating disorders, caregivers play a central role in helping their child recover. Specifically, a caregiver's self-efficacy, or their confidence in their ability to cope with the challenges their child faces during treatment, is believed to be a key to successful treatment outcomes. However, research on the impact caregiver self-efficacy has on patient treatment outcomes is mixed. Using the Parent vs. Eating Disorder (PVED) scale, we looked

*Correspondence: Dori M. Steinberg dori@equip.health Full list of author information is available at the end of the article



© The Author(s) 2023. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

at the association between caregiver self-efficacy and eating disorder symptoms and weight changes over time in a sample of 1051 patients receiving virtual FBT. Caregiver self-efficacy had little to no significant impact on treatment outcomes. We offer explanations about why our findings challenge existing research that supports the association between caregiver self-efficacy and outcomes. Namely, the PVED scale appears to measure caregiver perceptions of their role in the treatment process (and not self-efficacy). We believe a new measurement tool must be developed and tested for the field to move forward in its understanding of the role caregiver self-efficacy plays in FBT.

Introduction

Family-based treatment (FBT) is considered the goldstandard treatment for children and adolescents with an eating disorder. In randomized controlled trials, FBT has resulted in better remission outcomes than comparison treatments for adolescents with anorexia nervosa, bulimia nervosa, and, in a pilot study, avoidant restrictive food intake disorder (ARFID) [1-5]. Although FBT is a promising treatment, professionals still have limited knowledge about what predicts a successful outcome for children and adolescents receiving FBT. This is a critical concern given that only approximately 50% of youth fully recover after a course of FBT [6]. Family members are also an integral component of FBT, and clinicians often view parental empowerment as the most important part of FBT [7]. Indeed, caregiver or parental empowerment to bring about their child's recovery is one of the core tenants of FBT [8]. Therefore, it is also critical to understand parental factors associated with effective FBT, which has generally been overlooked to date.

In FBT, parents and/or other caregivers are included into the treatment team along with a trained psychologist or therapist, medical provider, and a dietitian [9]. It is the caregivers' responsibility to guide and supervise at-home aspects of treatment during the early stages of care. For example, a primary role of caregivers is to supervise and monitor meals and eating disorder behaviors, which supports the child's recovery [9]. FBT can also address the family burden of the illness by providing coping strategies to manage challenges related to their child's eating disorder [10]. The mealtime structure FBT provides, a pragmatic approach to symptom reduction, as well as support from the FBT therapist, helps caregivers increase their ability to cope with their child's eating disorder and improve familial relationships [11]. As progress is made in treatment, patients develop skills that enable them to take on more age-appropriate responsibilities and become increasingly independent.

Focus groups with parents involved in FBT to treat their child's eating disorder have found parents desire extra emotional support from treatment teams and eating disorder specific education that bolsters empowerment [12]. In the context of FBT, parental empowerment has been defined as building parental confidence to take on the role of primary agent of change in their child's recovery process, as building confidence allows caregivers to stand firm against their child's eating disorder [8]. A goal of parental empowerment, then, is to improve parent or caregiver self-efficacy in filling their important role in FBT, which may address caregiver concerns and improve treatment outcomes [13].

Self-efficacy refers to one's perceived ability or confidence to execute a specific behavior across a variety of situations [14]. Within FBT, parental self-efficacy is the caregivers' perceived ability to help their child recover from the eating disorder [13, 15]. Indeed, some studies have directly linked parental self-efficacy with better treatment outcomes [i.e., fewer eating disorder symptoms [13]] while others have specifically tied parental self-efficacy to greater weight gain [15, 16]. Such findings tying parental self-efficacy to positive treatment outcomes has resulted in increased attention to finding ways of supplementing FBT to bolster parental selfefficacy [17–19]. Multiple interventions to supplement FBT with a goal of improving parental self-efficacy have been studied, including providing structured parental psychoeducation, parental support groups, multifamily groups, and multifamily meals, among others [17-19]. Previous interventions designed to increase self-efficacy among parents in FBT have reported increased self-efficacy, yet parental self-efficacy was not associated with patient treatment outcomes [18, 20].

Other findings further complicate the understanding of parental self-efficacy as a predictor of treatment outcomes and suggest that the type of caregiver or parent (e.g., mother vs. father) may moderate the role of caregiver self-efficacy in treatment outcomes [16]. Some authors suggested that parental self-efficacy may influence treatment outcomes in FBT, yet not all studies support this assertion [21]. Hamadi and Holliday [21] discuss inconsistent findings in their review of factors that impact outcomes of adolescents receiving FBT treatment. The authors argue that, although parental self-efficacy has largely been accepted as a factor impacting treatment outcomes, the empirical literature is mixed (see [20–23], for further discussion and noted inconsistencies). Further, most studies evaluating parental self-efficacy to date have had a limited sample size, impacting statistical power, which in turn hinders the ability to detect reliable associations and test for moderators

Taken together, the aim of this study is to address several gaps in the literature to date. First, given the importance of caregivers in FBT, we evaluate the impact of parental factors on child treatment outcome. Second, we harness a large sample of young adults receiving FBT for an eating disorder. Specifically, we evaluate whether caregiver self-efficacy is associated with patient outcomes (i.e., eating disorder symptoms, weight) within the first 16-weeks of treatment.

Method

Participants and treatment overview

The sample for this study included patients (N=1051) and their caregivers self-selecting for eating disorder treatment at a national, virtually-delivered treatment program serving patients ages 6–24. Patients were either actively engaged in treatment or discharged from treatment with a diagnosis of anorexia nervosa, bulimia nervosa, binge-eating disorder, or other specified feeding and eating disorder, between the end of December 2020 and November 2022. Patients with ARFID were not included in this study.

The virtual treatment program delivers an enhanced version of FBT (FBT+) treatment of eating disorders. In FBT, a primary role of caregivers is to supervise and monitor meals and eating disorder behaviors, which support the child's recovery. In traditional FBT, treatment is led by a therapist and the team may include a registered dietitian and physician [9]. The FBT+approach includes a registered dietitian and physician as a standard member of the multidisciplinary treatment as well as a peer mentor and family mentor who have lived experience with an eating disorder. The peer mentor has recovered from an eating disorder and provides patients with a source of hope and motivation towards recovery. The family mentor has supported a loved one in recovery from an eating disorder and shares examples of skills and strategies with caregivers. If a family/patient chooses to not engage with a mentor, this is discussed with other members of the team in sessions but not forced; however, the mentor remains an integral member of the patients treatment team providing critical lived experience to other team members.

Our enhanced version of FBT is deployed in a way that fits the needs and specific presentation of the individual patient and family. Cadence of sessions is determined by the treatment team and as treatment progresses, tapering of sessions also occurs at the consensus of the treatment team and family generally occurring around the transition to Phase 3. All members of the care team meet with the family virtually through a Health Insurance Portability and Accountability Act (HIPAA) compliant platform approximately once per week. Following the traditional FBT model, FBT+ sessions are scheduled for 50-min sessions. More details about the FBT+ treatment approach and its effectiveness can be found elsewhere [22].

Measures

As a part of routine protocols, patients and caregivers completed empirically validated surveys during the course of treatment through a HIPAA compliant electronic health record.

Eating disorder symptoms

The Eating Disorder Examination Questionnaire Short Form (EDE-QS) was used to assess eating disorder severity [23]. This 12-item measure assesses eating psychopathology over the past seven days. Each item is rated on a four-point Likert-type scale ranging from 0 (no days) to 3 (six to seven days). Example items include "Have you had a definite fear that you might gain weight?" and "Have you tried to control your weight or shape by making yourself sick (vomit) or taking laxatives?" Scores can range from 0 to 36, with higher scores indicating more severe eating psychopathology. Scale reliability was estimated at Cronbach's a = 0.90 and EDE-QS measurements were collected weekly during treatment.

Weight

Patient weight was measured at home by a caregiver who was trained in weight monitoring procedures during orientation to FBT+. Weight was measured by trained caregivers twice weekly with the patient wearing minimal clothing, after voiding, and prior to food or beverage consumption. The electronic health record delivered an automated prompt to the caregiver to enter the weight via text or directly in the electronic health record.

Caregiver self-efficacy

The Parents Versus Eating Disorders (PVED) scale was completed by caregivers at admission and monthly thereafter. If patients have more than one primary caregiver involved in treatment, all caregivers are asked to complete the measure. The PVED is an adapted version of the Parents Versus Anorexia scale [24], which replaces the term "anorexia nervosa" with "eating disorder" to allow for use across various eating disorders. Previous studies exploring caregiver self-efficacy also use this minor change in wording to allow the scale to be applicable to experiences with any eating disorder [13, 17, 18, 25]. The Parents Versus Anorexia scale was developed for research of the Maudsley model of family therapy and designed to evaluate a caregiver's ability to take charge of the eating disorder in the home [24].

The PVED scale contains seven items measured on a five-point Likert-type scale ranging from zero (strongly disagree) to five (strongly agree). Example items include, "I feel equipped with specific practical strategies for the task of bringing about the complete recovery of my child in the home setting," "It is more my responsibility than my child's to bring him/her to a healthy weight" and "I don't have the knowledge to take a leadership role when it comes to achieving a total victory over the eating disorder." Total scores can range from seven to 35 with higher scores indicating greater perceived caregiver efficacy. Scale reliability was estimated at Cronbach's *a* = 0.61 in our sample, which is slightly lower than reported internal consistency during scale development [24].

Analytic strategy

We evaluated the impact of PVED scores on patient treatment outcomes during the first 16 weeks of FBT+. This time frame was chosen because it generally aligns with the end of treatment time frame in FBT randomized clinical trials [1]. We specified a series of multilevel models, as indicated below. The first model estimated PVED scores over time for caregivers (i.e., mothers and fathers). The model took the following form: $PVED_{w,j} \sim \beta_0 + \beta_{0j} + \beta_1 log(w) + \beta_{1j} log(w) + \beta_{1j} log(w)$ β_2 carergiver + β_3 [caregiver × log(*w*)]; where *w* represents the treatment week (we take the log because treatment progresses logarithmically over time [22]) and j indexes the patient (so that β_{0j} and β_{1j} represent random intercepts and slopes). The random effects account for different patient-level starting points and PVED change trajectories over time. The caregiver term represents the member of the patients FBT+support group that provided the PVED score in week w. This model outperformed a model that did not account for caregiver type (difference in expected log predictive density: 107.6 ± 14.9; difference in WAIC: 110.3 ± 15.0).

We next specified two multilevel models to estimate how eating disorder symptoms and weight (our primary outcomes) improved week-by-week through treatment, emphasizing the estimates of the association between PVED scores and these outcome trajectories. In weeks with more than one survey measurement, we averaged the measurement to obtain a single weekly value. Because we treat carers as a unit supporting the patient, we also averaged over caregiver PVED scores if they occurred in the same week.

Both models took the following form: $y_{wj} \sim \beta_0 + \beta_{0j} + \beta_1 \log(w) + \beta_1 \log(w) + \beta_2 PVED_0 + \beta_3 PVED_{\Delta} + \beta_4 [\log(w) \times PVED_0] + \beta_5 [\log(w) \times PVED_{\Delta}]$; where *w* represents the

treatment week, j indexes the patient (so that β_{0j} and β_{1j} represent random intercepts and slopes), PVED₀ represents the PVED score at week 0, and PVED_{Δ} represents the PVED change-from-baseline score in any given week *w*. The fixed effects portion of each model also included an interaction between the two PVED terms and treatment week.

This analytic approach allowed us to examine: (a) overall change in the treatment outcome over time, (b) separate the effects of the initial PVED score (a largely between subject variable) and the change in PVED over time (a largely within-patient variable), and (c) whether outcomes over treatment time varied as a function of PVED scores. We also included random intercepts and slopes in the model ($\beta_{0,j}$ and $\beta_{1,j}\log(w)$) to allow for individual differences in initial outcome values and individual trajectories through treatment. Given that the EDE-QS is an ordinal measure, we used an cumulative logistic link function (i.e., ordinal regression) to fit the model to these data [26]. Weight is a positive and an approximately continuous measurement, so it was fit with a standard gaussian link function.

The sample size for each model below differs slightly due to missing data. Even though both EDE-QS and weights were collected each week, patients/caregivers sometimes failed to complete the measurement. In some cases, carers failed to fill out the initial PVED survey, so we could not estimate PVED change scores for those families. Patients excluded from the analyses because of missing data did not differ demographically to those included.

We used Hamiltonian Monte Carlo to generate samples from the posterior distribution of the joint parameter distribution. This approach requires setting priors on the parameters to regularize the fits. We chose priors that fit the domain of the parameter, but otherwise kept the priors conservative. For example, our prior on log(w) (i.e., treatment week) for the EDE-QS outcomes model was (0, 2) even though we have prior data to show that we expect this parameter estimate to be negative, and around -2. It was set at (0, 6) in the weight outcomes model, even though we have prior data suggesting it should be closer to 3 or 4. Priors for all fixed effects covariates in the model were normally distributed priors centered around zero (null effect) but wide enough to capture larger effects. This results in a conservative test because the data has to overcome a higher prior probability density around the null hypothesis. The priors for the intercepts were chosen similarly: centered at zero for the EDE-QS ordinal model, and centered at 110 lbs for the weight model [based on [22]]. The priors on the random effects intercept and slope were half-(0, 4), to allow for large individual differences.

All analyses were conducted with R [27] using the tidyverse version 2.0 [28]. Fitting was done using the brms package version 2.18.0 [29], a wrapper around the Stan probabilistic programming language [30]. Model coefficients are presented with their corresponding 95% credible intervals. We consider a finding reliable if the credible intervals do not contain zero as a plausible value for the parameter. If zero is contained within the credible intervals, we consider the null hypothesis plausible. Project management was done using the targets package version v1.0.0 [31].

Results

Participants

Patients were on average 15 years-old (M = 15, SD = 2.43); patients were predominantly cisgender girls/women (n=862; 82.0%) and boys/men (n=85; 8.0%). Approximately 9.0% of patients (n=95) did not identify as cisgender and gender was unknown for nine patients. In all, 90.6% (n = 953) patients were diagnosed with anorexia nervosa, 3.4% (n = 36) with an unspecified eating or feeding disorder, 2.7% (n=29) diagnosed with bulimia nervosa, and 2.0% (n = 22) binge-eating disorder. In all, 46.0% of the patient sample completed at least 16 weeks of treatment. Throughout the treatment period under study, 174 patients were discharged before 16 weeks, and 131 patients had fewer than 16 weeks but were still active in treatment. The remainder of the sample remained active and completed 16 weeks of treatment. All patients with valid data were retained for further analysis, regardless of discharge status.

The sample included PVED ratings from 1,528 caregivers. Just over half of patients (53%) had at least two carers complete the PVED measure throughout treatment. Of these caregivers, 62.1% (*n*=949) self-identified as mothers (including step, adoptive, and foster mothers), 36.9% (n=565) self-identified as fathers, and the remainder reported other relationships with the patient (e.g., aunt, grandparent, significant other). At the onset of treatment, mothers reported an average PVED score of 20.58 (SD = 3.94), fathers reported 19.50 (SD = 3.56), and other caregiver types reported 20.00 (SD = 2.48); mothers reported slightly higher PVED scores than fathers $(M_{difference} = 1.08, t(1512) = 5.36, p < .001)$ but this may not be a clinically meaningful difference. Likewise, there was no difference in the rate of PVED change over treatment by caregiver type (b = -0.06, SE = 0.07, CI [-0.19, 0.07]). On average, across all caregivers, PVED scores significantly increased over treatment time (b=0.79, *SE*=0.04, CI [0.72, 0.86]; see Fig. 1).

Eating disorder symptoms

This model was fit using 10,596 observations across 895 patients. Eating disorder symptoms decreased over 16 weeks of treatment (b = -1.87, SE=0.10, CI [-2.06, -1.67]). The model term for starting PVED score estimated the between-patient difference in outcomes for a patient whose caregivers had higher initial PVED scores relative to the average score in our sample. Starting PVED scores were predictive of improved [decreased] eating disorder symptoms (b=-0.73, SE=0.22, CI [-1.15, -0.30]) such that at the onset of treatment, patients reported fewer eating disorder symptoms when their caregivers reported elevated PVED scores. Starting PVED scores were not moderated by treatment time (b=-0.02, SE=0.10, CI [-0.22, 0.18]).



Fig. 1 Predicted PVED scores over treatment time by caregiver type, and the corresponding weekly distribution of raw scores for each caregiver type

order symptoms (b = 0.03, SE = 0.03, CI [-0.03, 0.09]) and likewise did not moderate the association between treatment time and EDE-QS scores (b = .-0.02, SE = 0.02, CI [-0.05, 0.02]).

Weight

This model was fit using 15,873 observations across 863 patients. Weight increased over 16 weeks of treatment (b=3.47, SE=0.13, CI [3.20, 3.72]). Starting PVED scores were not significantly associated with weight (b=-0.96, SE=0.59, CI [-2.10, 0.19]). Likewise, the association between treatment time and weight did not depend on starting PVED scores (b=0.16, SE=0.12, CI [-0.08, 0.39]).

PVED change-from-baseline scores were predictive of weight (b = -.48, SE = .03, CI [-.53, -.43]) such that patient weight was lower when caregiver reports of PVED were higher. Likewise, the association between caregiver change in PVED scores and weight varied as a function of time in treatment (b = .27, SE = .01, CI [.24, .29]). Early in treatment (e.g., the first month), how a carer's PVED score changed had little bearing on weight. However, by 16 weeks of treatment, greater increases in PVED scores were associated with higher weight relative to similar PVED gains earlier in treatment (see Fig. 2).

Post-hoc analyses

Given that most studies of parental self-efficacy focus on individuals with anorexia nervosa, we also repeated the above analyses for eating disorder symptoms and weight including only patients with anorexia nervosa. The pattern of results did not change with only this subsample included in each analysis. Finally, we tested whether caregiver type (i.e., mother and father) moderated the relationship between PVED scores and each outcome. Across all models, caregiver type was not a significant moderator (EDE-QS: b = 0.01, SE = 0.01, CI [-0.02, 0.03]; weight: b = -0.01, SE = 0.01, CI [-0.02, 0.02]).

Discussion

Caregiver self-efficacy has been proposed to significantly impact patient eating disorder treatment outcomes during FBT [24], given that caregivers are an integral component to the treatment team [6]. However, the empirical evidence supporting this assertion is mixed [16]. In our large sample of patients receiving an enhanced version of FBT (FBT+), we observed a small or null association between caregiver self-efficacy and treatment outcomes. Further, although caregiver self-efficacy statistically moderated the association between time in treatment and weight change, the magnitude of this moderation was small and may not be clinically meaningful.

There are possible explanations for why we found small or null associations between caregiver self-efficacy and eating disorder treatment outcomes, as well as why the



research to date evaluating this has been mixed. First, the Parent Versus Anorexia questionnaire, which we modified to create the PVED, is one of the most commonly used measures to examine parental self-efficacy within the context of FBT. Yet, the internal consistency of this measure in our sample was low, and below the threshold generally considered adequate [32]. Low internal consistency aligns with other published and unpublished findings of the internal consistency of the PVA, indicating reliability is consistently low [22, 33]. Inconsistent reliability of the PVA/PVED is likely one important factor as to why results across studies in the literature are mixed. An unreliable measure is going to produce inconsistent results.

A review of the literature using the PVA scale reveals that the intended use of the measure may have also changed over time. As described by Rhodes and colleagues (2005), the PVA was developed as a measure of *parental efficacy*, defined by developers as "the ability of a parent to adopt a primary role in taking charge of the anorexia in the home setting for the purpose of bringing about the recovery of their child" [26]. Over the past decade, the description of this scale has transitioned to a measure of parental self-efficacy [16, 21, 33, 34]. Though nuanced, *efficacy* and *self-efficacy* are not entirely the same. Decades of research posits that self-efficacy involves a person's confidence in their ability to exert control over their motivation, behavior, and environment across different contexts [14]. Thus, standard measures of self-efficacy (across domains) frame items in a way that captures facets of one's confidence and belief in their ability to do a task or cope with challenge.

In contrast, most PVA items query parents/caregivers' perceptions of their role in FBT and how well parents/caregivers believe in the treatment process itself. For example, "I feel equipped with specific practical strategies for the task of bringing about the complete recovery of my child in the home setting" and "It is more my responsibility than my child's to bring him/her to a healthy weight." The items make sense given the PVA was designed to parallel the six core principles of the Maudsley model [24] and be consistent with the theoretical constructs of Maudsley. Taken together, the standard scale used in the literature to measure parental self-efficacy in FBT may not accurately represent how it was developed. A measure of parental self-efficacy during FBT should measure a caregiver's perceived confidence and ability to bring their child through eating disorder treatment. In contrast, the PVA (and the adapted version used in the current study; i.e., PVED) captures parental perceptions of their role in treatment and agreement with the FBT theoretical model-namely parental "buy-in" to FBT. Based on this, we can conclude that improving parental buy-in to FBT may not be enough to improve patient treatment outcomes.

Second, it is possible that PVED scores do not operate as we predicted. Here, we surmised that it operates consistently over time and thus we would find a systematic relationship with outcomes across time in treatment. However, PVED scores may function differently at different points during treatment. For example, others have shown an association between caregiver self-efficacy and treatment outcomes at 3 and 6 months after treatment [13]. Insofar as the impact of caregiver self-efficacy may vary with time, examining an overall association may not be the best fit to do the data. Other moderators may also exist such as family structure or disease burden. For example, nonintact families may experience greater illness burden [[35]; however see [36]] and as such, it is possible parental self-efficacy is more important for families reporting greater burden. Finally, our FBT + program provides additional support for parents not part of traditional FBT (e.g., family mentors). This consistent and enhanced family support, specifically the caregiver/ patient level impact of mentors, could impact findings in comparison to previous studies.

We take the transition of the conceptualization of the PVA scale over the past decade along with measurement challenges to be limitations of the current (and previous) work and potential factors that have contributed to the pattern of mixed findings in the literature. Although the PVA is widely used as a measure of caregiver self-efficacy, it was developed to measure a different, yet related, construct. It is imperative that future research construct and rigorously test a measure of caregiver self-efficacy that can be useful for answering questions about its role in the FBT process. Our study also has several strengths. We used a large clinical sample to study caregivers' perception of FBT and its association with eating disorder symptoms and weight. Moreover, we analyzed our data using advanced longitudinal modeling that takes into account the ordinal nature of severity of eating disorder symptoms data and provides estimates of the associations that are fit using prior knowledge (i.e., a Bayseian framework). Our analyses were also conducted with granular data (weekly outcomes measurements), which has been limitedly done previously.

Conclusions

Caregiver self-efficacy may be an important factor that bolsters desirable outcomes among patients in FBT for an eating disorder. Increasing a caregivers' belief in their ability to help the adolescent patient, as well as their confidence in providing needed support, makes logical sense as a factor that would improve treatment outcomes. However, in order to fully understand whether this is the case, a suitable measure of caregiver self-efficacy is needed. Understanding and measuring parental self-efficacy as a construct well grounded in self-efficacy research will open the door to new avenues of eating disorder research. Indeed, general self-efficacy measures have been used by eating disorder researchers in past studies [15, 16, 37] and may be an ideal starting point to develop a measure specific to caregivers of a child with an eating disorder. Furthermore, a new measure of parental self-efficacy in the eating disorder field will allow us to study the basic association between this construct and treatment outcomes; and over time include moderators, mediators, and nonlinear relationships. Separating selfefficacy from the other possible construct(s) that PVED measures allows us to better measure both the effects of self-efficacy and those other constructs (e.g., parental buy-in, or caregiver attitudes toward their role in treatment). Given the percentage of adolescents who do not fully recover from an eating disorder after a course of FBT, further understanding the importance of family factors in treatment may increase the likelihood of full recovery.

Abbreviations

 FBT
 Family-based treatment

 PVA
 Parent Versus Anorexia

 PVED
 Parent's versus eating disorders

 EDE-QS
 Eating Disorder Examination Questionnaire Short Form

Author contributions

NJ analyzed the data, drafted the manuscript, and conceptualized the project. JB assisted with drafting of the manuscript. DF assisted with data analysis, provided feedback on the manuscript, and conceptualized the project. BU assisted with drafting the manuscript and conceptualization. ACD conceptualized the project and provided feedback on manuscript drafts. CB conceptualized the project and provided feedback on manuscript drafts. DS conceptualized the project and provided feedback on manuscript drafts.

Funding

This project was funded by Equip Health, Inc.

Availability of data and materials

The data that support the findings of this study are derived from patient medical record data. The data are not publicly available and individual data cannot be shared due to privacy restrictions.

Declarations

Ethical approval and consent to participate

Evaluation of our existing treatments and patient outcomes was reviewed by the Western Institutional Review Board. It was determined such evaluations do not meet the definition of human subjects research and are therefore considered exempt from IRB oversight. Patients and/or caregivers provide informed consent during the admissions process for treatment information to be used for research purposes, including publication in scientific journals.

Competing interests

This project was funded by Equip Health, Inc. Authors are employees of Equip Health and may hold company stock options/equity.

Author details

¹Equip Health, Inc., CA, Carlsbad, USA. ²University of California, Irvine, CA, USA.
³Stanford University, Stanford, CA, USA. ⁴Duke University, Durham, NC, USA.
⁵Carlsbad, USA.

Received: 27 February 2023 Accepted: 11 August 2023 Published online: 22 September 2023

References

- Agras WS, Lock J, Brandt H, Bryson SW, Dodge E, Halmi KA, et al. Comparison of 2 family therapies for adolescent anorexia nervosa: a randomized parallel trial. JAMA Psychiatry. 2014;71(11):1279.
- Le Grange D, Crosby RD, Rathouz PJ, Leventhal BL. A randomized controlled comparison of family-based treatment and supportive psychotherapy for adolescent Bulimia Nervosa. Arch Gen Psychiatry. 2007;64(9):1049.
- Le Grange D, Lock J, Agras WS, Bryson SW, Jo B. Randomized clinical trial of family-based treatment and cognitive-behavioral therapy for adolescent Bulimia Nervosa. J Am Acad Child Adolesc Psychiatry. 2015;54(11):886-894e2.
- Lock J, Le Grange D, Agras WS, Moye A, Bryson SW, Jo B. Randomized clinical trial comparing family-based treatment with adolescent-focused individual therapy for adolescents with Anorexia Nervosa. Arch Gen Psychiatry. 2010;67(10):1025.
- Lock J, Robinson A, Sadeh-Sharvit S, Rosania K, Osipov L, Kirz N, et al. Applying family-based treatment (FBT) to three clinical presentations of avoidant/restrictive food intake disorder: similarities and differences from FBT for anorexia nervosa. Int J Eat Disord. 2019;52(4):439–46.
- Rienecke RD. Family-based treatment of eating disorders in adolescents: current insights. Adolesc Health Med Ther. 2017;8:69–79.
- Dimitropoulos G, Freeman VE, Lock J, Le Grange D. Clinician perspective on parental empowerment in family-based treatment for adolescent anorexia nervosa: parental empowerment. J Fam Ther. 2017;39(4):537–62.
- 8. Rienecke RD, Le Grange D. The five tenets of family-based treatment for adolescent eating disorders. J Eat Disord. 2022;10(1):60.
- 9. Lock J, Le Grange D. Treatment manual for anorexia nervosa: a familybased approach. 2., ed. paperback ed. New York: Guilford; 2015; 289.
- Treasure J, Nazar BP. Interventions for the carers of patients with eating disorders. Curr Psychiatry Rep. 2016;18(2):16.
- Wallis A, Rhodes P, Dawson L, Miskovic-Wheatley J, Madden S, Touyz S. Relational containment: exploring the effect of family-based treatment for anorexia on familial relationships. J Eat Disord. 2017;5(1):27.
- Cribben H, Macdonald P, Treasure J, Cini E, Nicholls D, Batchelor R, et al. The experiential perspectives of parents caring for a loved one with a restrictive eating disorder in the UK. BJPsych Open. 2021;7(6):e192.
- Robinson AL, Strahan E, Girz L, Wilson A, Boachie A. I know I can help you': parental self-efficacy predicts adolescent outcomes in family-based therapy for eating disorders: parental self-efficacy and adolescent outcomes in FBT. Eur Eat Disord Rev. 2013;21(2):108–14.
- Bandura A. Self-efficacy: toward a unifying theory of behavioral change. Adv Behav Res Ther. 1978;1(4):139–61.
- Byrne CE, Accurso EC, Arnow KD, Lock J, Le Grange D. An exploratory examination of patient and parental self-efficacy as predictors of weight gain in adolescents with anorexia nervosa: exploratory examination. Int J Eat Disord. 2015;48(7):883–8.
- Sadeh-Sharvit S, Arnow KD, Osipov L, Lock JD, Jo B, Pajarito S, et al. Are parental self-efficacy and family flexibility mediators of treatment for anorexia nervosa? Int J Eat Disord. 2018;51(3):275–80.
- Girz L, Lafrance Robinson A, Foroughe M, Jasper K, Boachie A. Adapting family-based therapy to a day hospital programme for adolescents with eating disorders: preliminary outcomes and trajectories of change: adapting family-based therapy. J Fam Ther. 2013;35:102–20.
- Spettigue W, Maras D, Obeid N, Henderson KA, Buchholz A, Gomez R, et al. A psycho-education intervention for parents of adolescents with eating disorders: a randomized controlled trial. Eat Disord. 2015;23(1):60–75.
- 19. Lock J, Le Grange D, Agras WS, Fitzpatrick KK, Jo B, Accurso E, et al. Can adaptive treatment improve outcomes in family-based therapy for

adolescents with anorexia nervosa? Feasibility and treatment effects of a multi-site treatment study. Behav Res Ther. 2015;73:90–5.

- Matthews A, Peterson CM, Peugh J, Mitan L. An intensive family-based treatment guided intervention for medically hospitalized youth with anorexia nervosa: parental self-efficacy and weight-related outcomes. Eur Eat Disord Rev. 2019;27(1):67–75.
- Hamadi L, Holliday J. Moderators and mediators of outcome in treatments for anorexia nervosa and bulimia nervosa in adolescents: a systematic review of randomized controlled trials. Int J Eat Disord. 2020;53(1):3–19.
- 22. Steinberg D, Perry T, Freestone D, Bohon C, Baker JH, Parks E. Effectiveness of delivering evidence-based eating disorder treatment via telemedicine for children, adolescents, and youth. Eat Disord. 2023;31(1):85–101.
- Gideon N, Hawkes N, Mond J, Saunders R, Tchanturia K, Serpell L. Development and psychometric validation of the EDE-QS, a 12 item short form of the eating disorder examination questionnaire (EDE-Q). PLoS ONE. 2016;11(5):e0152744.
- 24. Rhodes P, Baillie A, Brown J, Madden S. Parental efficacy in the familybased treatment of anorexia: preliminary development of the parents versus anorexia scale (PVA). Eur Eat Disord Rev. 2005;13(6):399–405.
- Lock J, Sadeh-Sharvit S, L'Insalata A. Feasibility of conducting a randomized clinical trial using family-based treatment for avoidant/restrictive food intake disorder. Int J Eat Disord. 2019;52(6):746–51.
- 26. Liddell TM, Kruschke JK. Analyzing ordinal data with metric models: what could possibly go wrong? J Exp Soc Psychol. 2018;79:328–48.
- 27. R Core Team. R: A Language and environment for statistical computing [Internet]. Vienna, Austria: R foundation for statistical computing. ; 2022. Available from: https://www.R-project.org/.
- Wickham H, Averick M, Bryan J, Chang W, McGowan L, François R, et al. Welcome to the Tidyverse. J Open Sour Softw. 2019;4(43):1686.
- Bürkner PC. Bayesian item response modeling in R with brms and Stan. J Stat Softw. 2021;100(5):1–54.
- 30. Stan Development team. RStan: The R interface to Stan [Internet]. 2022. Available from: https://mc-stan.org/.
- Landau W. The targets R package: a dynamic make-like function-oriented pipeline toolkit for reproducibility and high-performance computing. J Open Sour Softw. 2021;6(57):2959.
- Meyers LS, Gamst G, Guarino AJ. Applied multivariate research: design and interpretation. Third Edition. Los Angeles: SAGE; 2017. 978 p.
- Nash P, Renelli M, Stillar A, Streich B, Lafrance A. Long-term outcomes of a brief emotion-focused family therapy intervention for eating disorders across the lifespan: a mixed-methods study. Can J Couns Psychother. 2020;54(2):130–49.
- Le Grange D, Lock J, Agras WS, Moye A, Bryson SW, Jo B, et al. Moderators and mediators of remission in family-based treatment and adolescent focused therapy for anorexia nervosa. Behav Res Ther. 2012;50(2):85–92.
- Datta N, Hagan K, Bohon C, Stern M, Kim B, Matheson BE, et al. Predictors of family-based treatment for adolescent eating disorders: Do family or diagnostic factors matter? Int J Eat Disord. 2023 Feb;56(2):384–93
- Padierna A, Martín J, Aguirre U, González N, Muñoz P, Quintana JM. Burden of caregiving amongst family caregivers of patients with eating disorders. Soc Psychiatry Psychiatr Epidemiol. 2013;48(1):151–61.
- Pinto AM, Heinberg LJ, Coughlin JW, Fava JL, Guarda AS. The eating disorder recovery self-efficacy questionnaire (EDRSQ): change with treatment and prediction of outcome. Eat Behav. 2008;9(2):143–53.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

